Author's response to reviews

Title: Factors influencing identification of and response to intimate partner violence: a survey of physicians and nurses

Authors:

- Iris A Gutmanis (igutmani@uwo.ca)
- Charlene Beynon (cbeynon@uwo.ca)
- Leslie Tutty (tutty@ucalgary.ca)
- Nadine Wathen (nadine.wathen@utoronto.ca)
- Harriet MacMillan (macmilnh@mcmaster.ca)

Version: 3 Date: 23 January 2007

Author's response to reviews: see over
Re MS: 1812189519114007

We wish to re-submit our reformatted manuscript, entitled “Factors influencing identification of and response to intimate partner violence: a survey of physicians and nurses”, for publication as a research article in BMC Public Health.

Formatting changes have been made to the manuscript. As suggested, author affiliations have been changed. The word count has been removed. And, Table 3 is now a composite table and is not, as originally submitted, in sections. Finally, two commas have been deleted and three more added.

Thank you for your help with this publication.

Sincerely,

Iris Gutmanis

Telephone: 519-685-4292 ext. 42766
Fax: 519-685-4574
Email: igutmani@uwo.ca

cc: Charlene Beynon, Leslie Tutty, C. Nadine Wathen and Harriet L. MacMillan